Author's response to reviews

Title: Multiple Micro-nutrients Supplementation and Pregnancy Outcomes: A systematic review

Authors:

Tadesse Alemu Zerfu Mr (tadal2006@yahoo.com)
Henok Tadesse Ayele Mr (hennypho@yahoo.com)

Version: 4 Date: 19 November 2012

Author's response to reviews: see over
Dear Editor

This is the revised manuscript entitled “Multiple Micronutrients and Pregnancy: Effect of Supplementation on Pregnancy and Pregnancy Outcomes: A Systematic Review.” We thoroughly gone through the comments of the three reviewers and we almost revised the manuscript according to their comments. Starting from the topic, then the problem of statement, methodological description, result & discussion and references were the main areas of reviewers interest and we made a revision according to their comments. Because of academic reason me and my coauthor are living in different countries that made extended delay of revision. I am gatefull for all the patience to this end. I hope the next revision if any will not be as delayed as this one. Hope you will reconsider the systematic review for the publication.

Regards!

Henok T. Ayele
UMC, The Netherlands!